Author's response to reviews

Title: HER2 testing on core needle biopsy specimens from primary breast cancers: interobserver reproducibility and concordance with surgically resected specimens

Authors:

Hitoshi Tsuda (hstsuda@ncc.go.jp)
Masafumi Kurosumi (mkurosumi@cancer-c.pref.saitama.jp)
Shinobu Umemura (umemura@is.icc.u-tokai.ac.jp)
Sohei Yamamoto (nrg09280@nifty.com)
Takayuki Kobayashi (dr20010@ndmc.ac.jp)
Robert Yoshiyuki Osamura (osamura@is.icc.u-tokai.ac.jp)

Version: 4 Date: 26 September 2010

Author's response to reviews:

Dear Editor:

We are pleased to be informed that the manuscript is suitable for publication in BMC Cancer, pending formatting revisions.

According to your comments, we have amended the format. Authors' list, Table 3, Table 4, and Table 7 have been amended according to the comments.

For minor revisions, text, abstract, introduction, and materials and methods have been amended. For figures, titles have been removed, and the space around the image has been minimized.

Several typographical errors have been amended.

If there are insufficient points in the amendments, please let me know.

Sincerely yours,

Hitoshi Tsuda